Author’s response to reviews

Title: The effect of menopause and hysterectomy on systemic vascular endothelial growth factor in women undergoing surgery for breast cancer.

Authors:

Aoife J Lowery (aoife.lowery@gmail.com)
Karl J Sweeney (karlj.sweeney@hse.ie)
Alan P Molloy (alanpmolloy@yahoo.com)
Emer Hennessy (emer.hennessy@nuigalway.ie)
Catherine Curran (Catherine.Curran@hse.ie)
Michael J Kerin (michael.kerin@nuigalway.ie)

Version: 7 Date: 30 September 2008

Author’s response to reviews:

Department of Surgery
Clinical Science Institute
University College Hospital
National University of Ireland Galway
Ireland
` 30-09-08

Sabina Alam, PhD
Assistant Editor
BioMed Central


Dear Dr. Alam,

Many thanks for reviewing this manuscript for publication in BMC Cancer

The formatting changes have been made to the paper as requested:

Highlighting/tracking- the tracking has been removed from the manuscript. To our knowledge there is no highlighting in the manuscript.

Figure as two files - Figure 2 has been reformatted to create a single composite figure with the sub-labels A and B.

We hope that these changes are in keeping with the requirements for the journal.
Sincerely,

Professor Michael Kerin
Department of Surgery
Clinical Science Institute
University College Hospital
National University of Ireland Galway
Ireland
e-mail: michael.kerin@nuigalway.ie
Tel: + 353 91 524390
Fax: + 353 91 494509